Author's response to reviews

Title: Spectrum and antibiotic sensitivity of bacteria contaminating upper gut in patients with malabsorption syndrome from the tropics

Authors:  
- Dr Ujjala Ghoshal (ujjala@sgpgi.ac.in)  
- Uday C Ghoshal (ghoshal@sgpgi.ac.in)  
- Piyush Ranjan (pranjan@sgpgi.ac.in)  
- Subhash R Naik (ghoshal@sgpgi.ac.in)  
- Archana Ayyagari (archana@sgpgi.ac.in)

Version: 5 Date: 18 May 2003

Dear Sir,  
Kindly find revised manuscript. Major changes made are:  
1. Grammar and spelling mistakes corrected.  
2. References made up to date.  
3. Fig. 1 cropped. Fig. title removed.  
4. Authors affiliation checked.  
5. Pink fonts are made black.  
6. Table 1 and 2 placed within manuscript.  
Thanks.  
Uday Ghoshal